Author's response to reviews

Title: Can primary care team-based transition to insulin improve outcomes in adults with type 2 diabetes? The Stepping Up to Insulin cluster randomised controlled trial protocol

Authors:

John S Furler (j.furler@unimelb.edu.au)
Doris Young (d.young@unimelb.edu.au)
James Best (jdbest@unimelb.edu.au)
Elizabeth Patterson (elizabeth.patterson@unimelb.edu.au)
David O'Neal (dno@unimelb.edu.au)
Danny Liew (Danny.Liew@mh.org.au)
Jane Speight (JSpeight@acbrd.org.au)
Leonie Segal (Leonie.Segal@unisa.com.au)
Carl May (c.r.may@soton.ac.uk)
Jo-Anne Manski-Nakervis (jomn@unimelb.edu.au)
Elizabeth Holmes-Truscott (ETruscott@acbrd.org.au)
Louise Ginnivan (louiseeg@unimelb.edu.au)
Irene Blackberry (i.blackberry@unimelb.edu.au)

Version: 4 Date: 6 February 2014

Author's response to reviews:

I will upload a clean version now

I rejected only one change where you shortened practice nurse to PN on the 1st occasion it was used ie needed spelling out in full on that occasion (se first para of abstract)